Contents lists available at ScienceDirect







journal homepage: www.elsevier.com/locate/esr

Causality between residential electricity consumption and explanatory factors

Serge Guefano ^{a,*}, Mokhtar Bozorg ^a, Mauro Carpita ^a, Pauline Darline Etet-Baha ^b, Francis Lénine Djanna Koffi ^c, Jean Gaston Tamba ^c, Jacques Etame ^d

^a Energy Institute, School of Management and Engineering Vaud, University of Applied Sciences Western Switzerland (HES-SO), Yverdon-les-Bains, Switzerland

^b University Institute of Technology, University of Douala, Douala, Cameroon

^c Laboratory of Technology and Applied Sciences, P.O BOX 8698, IUT Douala, Douala, Cameroon

^d University of Douala, Department of Earth Sciences, Douala, Cameroon

ARTICLE INFO

Keywords: Electricity consumption Causal relationship Explanatory factors Sustainable development Residential sector Cameroon

ABSTRACT

This research adds to the assessment of the causal relationships between residential electricity consumption and its explanatory factors. Johansen cointegration, autoregressive distributed lag model, the vector error correction model, and Wald causality tests are used. In both the short and long run, a unidirectional causality towards electricity consumption is identified. The sustainable development of the electricity sector in Cameroon, as defined by the national development strategy for 2030 and the long-term electricity sector development plan for 2030, requires an increase in GDP per capita, the adoption of a compact less energy-intensive urban model, the strengthening of control over the evolution of the number of subscribers on the distribution networks, and the reduction of CO2 emissions from thermal electricity generation.

1. Introduction

Modern energy services are essential to the long-term well-being of populations and to global sustainable development [1]. Access to clean energy is a goal shared by both developed and developing countries. In this case, access to electricity for households, healthcare facilities, schools and public services are shared priorities at all levels of development.

In the category of developing countries working to improve energy services, we find Cameroon, where 71% of households consume electricity on low voltage (LV), corresponding to an electrical voltage that varies between 220 V and 240 V [2]. A survey conducted in December 2018 grouped households according to electricity power consumption ranges, namely: 2.2 to 2.4 kW, 3.3 to 3.6 kW, and 6.6 to 7.2 kW, with current intensities of 10 Amperes (A), 15 A and 30 A, respectively [2,3]. There are also around 29% of small and medium-sized enterprises (SMEs) as well as small and medium-sized industries (SMIs) that consume electricity on medium voltage (MV) [4,5]. These companies take out a subscription with a power of between 0.036 MW and 4 MW, for a voltage range of between 3 kV and 33 kV [6].

Access to electricity for Cameroonian households differs according to the level of the local population [7]. In urban areas, for example, the rate of access to electricity is 93.19% for a population of more than 14 million inhabitants in 2019 [8]. The rate of access to this energy is

estimated at 94.03% for a population estimated at over 15 million in 2020 [8]. In addition, the rate of access to electricity in rural areas is estimated at 24.18% for a population of 11,094,657 in 2019. In 2020, the rural population is estimated at 11,242,817, for an electricity access rate of 24.98% [8]. Over these two years, the rate of access to electricity increases as the population grows. However, the difference in access rates between rural and urban areas clearly shows that more needs to be done to improve electricity supply to the least satisfied sections of the population. With a 7% share of Cameroon's energy mix in 2016 (Fig. 1), electricity is the third most important energy source used by Cameroonian households, after wood energy and liquefied petroleum gas (LPG) [9]. Electricity consumption in households is mainly for lighting and powering household appliances, while the end uses of electricity in SMEs and SMIs include lighting and the production of goods. Given these end uses in the residential sector, electricity appears to be a clean energy whose production and supply in sufficient quality and quantity should be increased in order to bring the residential sector into line with sustainable development objectives.

Cameroon has one of the greatest hydroelectric potentials in sub-Saharan Africa (19.7 GW), of which only 5% is used for electricity generation [10]. To improve this situation, over the last two decades the government has approved various policies aimed at increasing

* Corresponding author. E-mail address: serge.guefano@heig-vd.ch (S. Guefano).

https://doi.org/10.1016/j.esr.2023.101155

Received 5 December 2022; Received in revised form 23 July 2023; Accepted 3 August 2023 Available online 17 August 2023

2211-467X/© 2023 The Author(s). Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).



Fig. 1. Percentages of energy use in Cameroon.

its electricity generation capacity. These include the development of public-private partnerships, the diversification of projects aimed at increasing the number of electricity generation, transmission and distribution infrastructures nationwide, and the readjustment of laws regulating the activities of the various stakeholders in Cameroon's electricity sector [11]. Despite these many efforts, the quality of electricity supply remains mediocre. Consumers in Cameroon's residential sector are frequently subjected to untimely power cuts, which tarnish social well-being [3]. The gap between the rates of access to electricity in rural and urban areas remains wide in 2020. These rates are estimated at 94.03% in urban areas, compared with 24.98% in rural areas on the aforementioned date [8]. In view of these figures, the sustainable development objective, which in sub-section 7-1 supports access to available electrical energy for all households [12], is still a pipe dream in the Cameroonian context. As a result, electricity use in the residential sector seems to be linked to a number of factors that are holding back the achievement of sustainable development objectives. Consequently, this study aims to identify various economic, demographic and climatic factors that influence electricity consumption in the residential sector, and to highlight the causal relationships that exist between these factors and such consumption, with a view to suggesting development orientations that could contribute to the long run development of the electricity sector in Cameroon.

1.1. Literature review

Analyses of the causal links involving electricity consumption and various influencing factors give rise to policy recommendations capable of contributing effectively to the sustainable development of the various sectors of electrical energy consumption in Africa. In 2019, for example, [13] assessed the causal link between electricity use in Zimbabwe, gross domestic product per capita (GDPC) and CO2 emissions over the period 1971-2014. This analysis reveals the existence of a unidirectional causal relationship. Electricity consumption drives Zimbabwe's economic growth and leads to carbon dioxide emissions. The energy policies formulated recommend increasing electricity generation using various sources of renewable energy that are conducive to preserving the environment. Such an initiative should help the country to develop its economy while reducing its carbon dioxide emissions. [14] studied the relationship between renewable energy consumption and economic growth in 16 emerging countries, including Egypt and South Africa. No causal relationship was found between energy consumption and economic growth in these two African countries. This underlines the opportunity to develop energy efficiency policies without slowing down the country's economic growth. [15] studied the long run relationship

between electricity consumption, public spending and sustainable development in Nigeria. The study reveals a negative impact of recurrent government spending on GDPC and electricity consumption. The policy recommendations arising from these observations emphasise the need to take account of real per capita needs when implementing development strategies and projects aimed at improving per capita electricity supply.

[16] investigated the existence of a causal relationship between electricity consumption (EC), economic growth and trade openness (TRO) of South Africa. Granger causality tests highlight the existence of a unidirectional causal relationship from electricity consumption to the country's GDP and trade openness, respectively. The energy policies highlighted suggest an increase in electricity generation in order to support the country's trade openness and economic growth. In addition, [17] examined the possible interactions between electricity consumption, CO2 emissions and economic growth in Cameroon. The investigations carried out show that there is no interaction between electricity consumption and economic growth, and that CO2 emissions are unidirectionally linked to economic growth. Electricity generation therefore appears to be insufficient to support the country's economic development, while fossil fuel consumption linked to generation activities sustains CO2 emissions on the one hand, and promotes the country's economic development on the other. The energy strategies formulated recommend increasing hydroelectric generation to support economic growth. In 2020, studies highlighting energy policies derived from causal relationship analyses involving electricity consumption in Africa are available. In the Nigerian context, [18] identifies a unidirectional causality from electricity consumption (EC) to urbanisation (UR) and economic growth, respectively. Policy recommendations encourage the enrichment of the energy mix to meet the growing energy demand in urban areas. [19] identifies a unidirectional causality from electricity consumption to South African economic growth. Electricity consumption with an energy intensity of less than 0.48% stimulates the country's economic growth.

1.2. Research gap and contributions

The analyses presented above demonstrate the relevance of the energy strategies proposed following studies of the causal relationships between electricity consumption and various factors interacting with this consumption in different African economies. However, very little research has focused on studying these interactions within the various African residential sectors. Table 1 summarises the results of causality studies involving electricity consumption in Africa over the timeframe 2011–2020. Based on these investigations, two studies were identified

Energy Strategy Reviews 49 (2023) 101155

Summary of	main causanty studies niv	orving cicci	ficity consumption in Africa.		
Author	Country	Year	Methodologies	Period	Causality
[22]	Ghana	2011	Toda and yamamoto; Granger causality	1971-2008	$GDP \rightarrow EC$
[23]	Ivory Coast	2011	ARDL; VECM	1980-2008	$ECC \leftrightarrow GDPC$
[24]	Botswana	2011	ARDL	1980-2008	$EC \rightarrow GDP$
[25]	Nigeria	2012	ARDL-VECM	1970-2008	$EC \rightarrow CO2$
[26]	Nigeria	2013	VECM	1970-2005	$GDP \rightarrow EC$
[27]	Nigeria	2013	Cobb–Douglas model; VECM	1980-2008	ECC \leftrightarrow GDP
[28]	Algeria	2013	VECM	1971-2010	$EC \leftrightarrow GDP$
[29]	Angola	2013	VECM	1971-2009	$EC \leftrightarrow GDP; \ EC \leftrightarrow UR$
[30]	Egypt, South Africa	2014	Panel Granger causality	1970-2011	$GDP \leftrightarrow EC$
[31]	24 African countries	2014	Granger causality	1971-2009	$EC \leftrightarrow UR$
[32]	Nigeria	2015	VECM	1971-2011	$EC \rightarrow GDP$
[33]	South Africa	2015	Granger causality	1971-2009	ECC-no causality-GDPC
[34]	Nigeria	2015	ECM; Granger causality	1970-2012	$EC \leftrightarrow GDP;EC \leftrightarrow EP$
[35]	Ivory coast, Zambia	2015	VECM; Granger causality	1971-2011	FID \rightarrow EC; GDP \rightarrow EC
[36]	Ghana	2016	VECM; Granger causality	1970-2014	$GDP \rightarrow EC$
[37]	Uganda	2016	VECM; Granger causality	2005-2016	$EC \rightarrow GDP$
[38]	South Africa	2017	ARDL; Toda and Yamamoto	1971-2012	$CO2 \rightarrow EC$
[39]	Ghana	2018	Toda and Yamamoto	1971-2014	$EC \rightarrow IG$
[40]	Egypt	2018	VECM; Toda and Yamamoto	1971-2013	$EC \leftrightarrow GDP$
[20]	Algeria	2018	ARDL	1970-2013	$GDP \rightarrow REC$
[13]	Zimbabwe	2019	Toda and Yamamoto	1971-2014	$EC \rightarrow GDP$
[16]	South Africa	2019	Granger causality	1984-2015	
[18]	Nigeria	2020	FMOLS; DOLS; CCR	-	$EC \rightarrow GDP$
[21]	Tunisia	2020	Granger Causality	1980-2018	REC \rightarrow EP; REC \rightarrow UR

of main according studies involving alastnisity consumption in Africa

Notes : \rightarrow : unidirectional, \leftrightarrow : bidirectional, GDP: Gross Domestic Product, EC : Electricity Consumption, ECC : Electricity Consumption per Capita, GDPC : Gross Domestic Product per Capita, UR : Urbanisation, EP : Electricity Price, FDI: Foreign Direct Investment, CO2 : carbon dioxide; IG: Industrial Growth; REC : Residential Electricity Consumption, ECM : Error Correction Model, VECM : Vector Error Correction Model, ARDL : Autoregressive Distributed Lag, FMOLS : Fully-Modified Ordinary Least Squares, DOLS : Dynamic Ordinary Least squares.

as being interested in the analysis of causal relationships involving electricity consumption in the residential sector. In 2018, [20] studied the causal relationships between electricity consumption in the Algerian residential sector and its GDP. The study highlights the existence of a unidirectional causality from GDP to electricity consumption. An increase in per capita income would lead to an increase in the number of household appliances and higher electricity consumption in the residential sector. For this reason, energy efficiency measures are proposed, in particular the use of less energy-consuming appliances. In addition, it is recommended that the use of available renewable energies be stepped up in order to meet the growing demand for electricity driven by the country's economic growth. In 2020, [21] analyses the demand for electricity in the Tunisian residential sector using three explanatory factors: GDP, the price of electricity (EP) and urbanisation (UR). It is shown that there is a unidirectional causality from residential electricity consumption (REC) to the price of electricity and urbanisation, respectively. summarises the main directions and senses of causality identified. Unfortunately, this second study, which focuses on electricity consumption in the residential sector in Africa, fails to formulate energy strategies that will make it possible to meet the growing demand for electricity in the face of increasing urbanisation, and to adjust the price of electricity to consumers' wallets. Consequently, this work is a continuation of the two previous studies, with the aim of analysing, on the one hand, the causal relationships between electricity consumption in the Cameroonian residential sector and its main influencing factors, and on the other hand, the causal relationships identified should make it possible to develop energy strategies that should contribute to the sustainable development of the electricity sector in the Cameroonian residential sector.

1.3. Paper organisation

The remainder of this work is organised as follows: Section 2 details the theoretical framework of the study, Section 3 describes the data relating to the main factors which have a significant impact on residential electricity consumption, and which have been selected with regard to the conceptual framework described and similar studies. The econometric methodology used to identify the direction and sense of

causal relationships between these influencing factors and electricity consumption is described in Section 4. Section 5 presents the results, Section 6 the main discussions, and Section 7 sets out the main conclusions and suggested energy strategies following the analyses of causal relationships discussed in the previous sections.

2. Theoretical framework of the study

Over the period 2010-2020, there are only two studies involved in analysing electricity consumption and its various influencing factors in Cameroon. In 2017, Tamba et al. [41]. conducted the first study, assessing the existence of a causal relationship between national electricity consumption and economic growth modelled by its GDP over the period 1971-2013. The econometric approach applied in this study combines stationarity tests, the Johansen cointegration test, the Granger causality test and vector autoregressive modelling (VAR model). The absence of cointegration and causality between the two variables is highlighted. National electricity generation up to 2013 appears to be insufficient to sustain economic growth. The energy strategies formulated are based on exploiting the strong existing potential, i.e. 294 Terrawatt-hours (TWh) exploited at 4.15% until 2013. The second study is scheduled for 2019. Njoke et al. [17]. carry out an empirical analysis of existing relationships between national electricity consumption, CO2 emissions and economic growth over the period 1971-2014. The econometric theory used reveals multifaceted interactions between the variables. ARDL (autoregressive distributed lag model) modelling and upper bound tests highlight the existence of long run cointegration between CO2 emissions and economic growth. The Toda and Yamamoto Granger causality test reveals the existence of unidirectional causality from CO2 emissions to economic growth. In addition, the Kuznets curve hypothesis reveals the existence of an inverted U-shaped curve between economic growth and the increase in CO2 emissions. The hypothesis of neutrality between national electricity consumption and economic growth is also supported by this second study. The energy strategies formulated recommend improving electricity generation using renewable energy sources, with the corollary of improving electricity access rates, estimated respectively at 94.03% and 24.98% in urban and rural areas in 2020 [8], and reducing CO2 emissions from thermal

electricity generation, estimated at 6550 kt of CO2 in 2019 [1]. Due to the unavailability at national level of a wide range of balanced data for all the variables under study, the data collected generally comes from international institutions such as the World Bank (World Bank's development indicators) [17,41].

Until 2022, no study has analysed the interactions between sectoral electricity consumption and various influencing factors in Cameroon. However, during 2022, Tamba et al. [2] make an overview analysis of the Cameroonian electricity sector and identify factors that may interact with electricity consumption. These factors are called factors influencing electricity consumption in Cameroon, and are grouped into six main categories presented in this study under the label ($(\alpha_i)_{1 \le i \le 6}$).

 (α_1) : Socio-economic factors have the particularity of increasing electricity consumption in the residential sector. These include GDP, GDP per capita, the number of households and the number of subscribers. It would therefore be interesting to verify this hypothesis on the basis of a structured econometric theory.

 (α_2) : geographical factors are the second category presented, with the particularity of influencing the country's hydrology and national electricity generation, 61.7% of which will be supported by hydroelectricity in 2019 [1]. With dams capable of storing 99.92% of cubic metres of water during periods of high water, and only 6.12% of their capacity during periods of low water, electricity generation is subject to strong fluctuations. The consequences of this situation include instability of the electricity grids and frequent household blackouts [2].

 (α_3) : The category of end users includes consumers in the residential sector, made up of households, SMEs and SMIs. In 2020, consumption reached the scale of 1470.3 Gigawatt-hours (GWh), behind that of the industrial sector, which is also listed in this category, and whose electricity consumption is high at 3845.8 (GWh), compared to 2019 when this consumption stood at 3618.3 GWh [1]. With regard to hydroelectricity generation in Cameroon, estimated at 5229 GWh in 2019 and 5434 (GWh) in 2020 respectively [1], it can be seen that electricity consumption in the industrial sector corresponds to 69.2% and 66.6% of previous generation between these two years. The hydrological disturbances described above, together with the frequent power cuts they cause, have a negative effect on industry in Cameroon. The average cost of an interruption varies between €3.62/kWh and €5.42/kWh for a one-hour outage [42], and between €1.96/kWh and €2.46/kWh for a 4-hour power cut [42]. This situation is undermining the country's industrial generation and economic growth.

 (α_A) : Climatic factors capable of significantly influencing electricity consumption in the residential sector include meteorological factors such as rainfall, temperature and humidity [43]. Cameroon's climate is characterised by three main seasons [2]. (1): The dry season, from November, December, January and February, with rainfall of less than 200 mm/month and temperatures between 24.5 °C and 26 °C/month. (2): The short rainy season, which extends over April, May, June and October, with rainfall of between 200 mm and 400 mm/month, and average temperatures hovering around 25 °C/month. (3): the long rainy season, from July, August, September to May, including the first week of October. The corresponding rainfall is over 400 mm/month, with temperatures between 23.5 $^{\circ}\text{C}$ and 24 $^{\circ}\text{C/month}.$ Depending on the variations in these different meteorological factors, there are different peaks in electricity consumption on the two major existing electricity networks. On the southern interconnected network (SIN), which serves 6/10 of the existing regions (the South, Centre, Coast, West, North-West and South-West regions), electricity consumption peaks are recorded, fluctuating between 475 MWh/month and 570 MWh/month [43]. On the Northern Interconnected Network (NIN), which covers 3/10 of the country's regions (the North, Adamaoua and Extreme North), peak electricity consumption fluctuates between 35 MWh/month and 45 MWh/month [43].

 (α_5) : the category of technical factors includes outdated electricity networks, uncontrolled and fraudulent interconnections to distribution networks, and major losses on these networks. The dilapidated state

of the SIN and NIN leads to significant losses during the transmission and distribution phases. For national electricity generations of 8476 GWh and 9008 GWh in 2019 and 2020 [1], electricity losses are 1864 GWh and 1981 GWh respectively [1]. Around 21.99% of the electricity produced is lost on the electricity networks before it reaches the end consumer. In addition, losses due to fraudulent connections, fires, power cuts and network maintenance are estimated at 914 GWh and 971 GWh [1], i.e. 10.78% of national electricity generation in 2019 and 2020 respectively. It is therefore urgent that the State of Cameroon adopt energy strategies to reduce these losses to a minimum, and contribute more effectively to the long run development of the electricity sector.

 (α_6) : This last category of influencing factors concerns political factors [2]. These are all the strategies, decisions and projects contributing to the development of the electricity sector in the short and long run. Cameroon's electricity sector development projects include the national development strategy 2020-2030 (SND30), and the long run electricity sector development plan 2030 (PDSE-2030). In the SND30 [44], the government presents a reference framework to support multi-sector development up to 2030. In the electricity sector, the government is committed to reducing the gap between supply and demand by increasing installed electricity capacity to 5000 MW by 2030. This objective involves the construction of around ten hydroelectric power stations nationwide, with a combined installed capacity of 5226 MW. The government also plans to subsidise the construction of mini hydroelectric power stations in localities far from the main transmission and distribution networks, in order to meet household demand in those areas.

The PDSE-2030 provides for the extension of power lines and the interconnection of the SIN and NIN to form the National Interconnected Network (NI) by 2030 [45]. In this theoretical context, it is important to develop energy strategies that will contribute to the sectoral development of electricity over the long run. In the context of the residential sector, electricity consumption is considered as a proxy for electricity demand. The factors that influence this consumption over the short and long run are referred to as electricity consumption explanatory factors, following the example of several previous studies [46–49].

3. Data

This section describes the data relating to the various variables under study and their respective sources. These are mainly electricity consumption in the residential sector, GDP per capita, national CO2 emissions, urbanisation and the number of electricity-consuming subscribers on the medium-voltage and low-voltage networks. Research similar to this study recommends including the price factor in the analysis of causal relationships involving electricity consumption, in order to capture its effect on demand trends. However, the interval distribution of this factor makes it difficult to trace its evolution over previous years, and hence its inclusion in this causal analysis. This aspect is discussed in more detail in this section. Since the variable to be explained is electricity consumption, we also describe a number of tests to justify the statistical relevance of the explanatory factors selected.

3.1. Data description

In order to contribute to the enrichment of studies of causal relations linked to electricity consumption in the residential sector, four explanatory factors with an annual frequency are used. These include GDP per capita, an indicator of individual income, and the ability of each person to take out a continuous electricity subscription [50]. CO2 emissions are listed among the factors having a causal relationship with electricity consumption [51,52]. Although [17] formulates the hypothesis of an absence of causality between CO2 emissions and national electricity consumption in Cameroon, it would be interesting to carry out a targeted assessment within the residential sector. Following the



Fig. 2. Trends in specific variables.

analyses of Talbi et al. [21], we also retain the urbanisation factor within the framework of this study. However, the price factor is not taken into account in this study. Prices per kWh are stable and depend only on consumption ranges. For consumption below 110 kWh, the price is set at USD 0.09/kWh, between 111 kWh and 400 kWh, the price is USD 0.145/kWh, between 401 kWh and 800 kWh, the price is uSD 0.170/kWh, while between 801 kWh and 2000 kWh, the price is set at USD 0.180/kWh [53]. This segmented distribution of electricity prices within the residential sector makes it difficult to take account of any price trends that might be compared with changes in electricity consumption. The number of subscribers is also included in the list of factors explaining electricity consumption. This is based on the assumption that there is a fairly close link with electricity consumption [54,55] Fig. 2 shows a similar trend for the variables selected, suggesting a very close interrelationship between their statistical indicators.

3.2. Data sources and statistical consistency

The databases of the International Energy Agency [1] provide information on electricity consumption in Cameroon's residential sector (REC in GWh). Data on GDP (in current US dollars), CO2 emissions from thermal electricity generation (in kilotonnes of CO2) and urban population (UR), are available in the World Bank database [8]. Finally, data on the number of subscribers (NS) are provided by the electricity distributor ENEO Cameroon [56]. These variables cover the period 1994-2019, during which the country's economy and demographic factors are growing overall. Table 2 shows some descriptive statistics based on these data. The mean, maximum and minimum values, standard deviation, skewness and kurtosis are all included. If the skewness of a given time series is in the range [-3;3] and its kurtosis is in the range [-10; 10], then the series has a normal distribution [57]. In this context, the time series is representative of the physical process it describes [57]. Table 2 shows that the kurtosis and skewness of each time series meet the above reliability conditions. Consequently, the associated data are reliable representations of the real fluctuations of each variable.

These factors are among the indicators that best explain residential electricity consumption in Cameroon. Table 3 describes the strong correlation between REC and each explanatory factor, with correlation coefficients generally fluctuating around 1. Analysis of the relevance of the explanatory factors using the ordinary least squares method shows that they have t-statistics in absolute values above the reference of

1.96 and P-values overall below the 5% threshold. These factors are therefore 95% statistically significant and suitable for justifying the temporal variations in electricity consumption.

Fig. 3 shows the trends in growth rates for the five variables studied over the period 1995–2019. The very close link between the different trends confirms the results of the Table 3, which show a strong correlation between residential electricity consumption and each explanatory factor.

Table 4 confirms the hypothesis of a very close correlation between the variables. Most of the correlations between the growth rates are significant at the 1% level.

The steps involved in analysing the causal link between residential electricity consumption and each explanatory factor are described in this section. During these steps, statistical methods are used to figure out the best number of lags, the existence of cointegration and the direction of causality in the short and long run, respectively.

4. Empirical methodology

This section briefly describes the statistical tests used sequentially to model electricity consumption in the residential sector as a function of its determinants. These include the stationarity tests process (Section 4.1), the criteria for selecting the optimal lag number for the variables (Section 4.2), and the Johansen cointegration test process for determining the order of long-term cointegration of the variables in the model (Section 4.3). The Johansen test carried out here is based on variables relating to 26 annual observations, whereas the Johansen test generally requires a wider range of observations to be truly relevant. For this reason, Section 4.4 presents Autoregressive Distributed Lag (ARDL) technique developed to validate the results of the Johansen test. Section 4.5 presents the Vector Error Correction Model (VECM) used to assess the direction of the causal relationship between the different variables, in the short and long term, respectively. Finally, Section 4.6 describes some statistical tests to validate the robustness of the specified VECM model.

4.1. Stationarity tests process

The raw data for the majority of time series variables are nonstationary time series [58]. To avoid any biased causality analysis, it is necessary to ensure their stationary properties [59]. To do this, two statistical tests are used in this study: the Augmented Dickey Fuller

Descriptive statistics of variables

···· I · · · · · · · · · · · ·					
Descriptive statistics	REC	GDPC	CO2	NS	UR
Mean	782.2692	1089.2980	7392.3080	653.3980	9509759
Standard Deviation	391.4480	334.2153	906.2066	249.4699	2812682
kurtosis	1.5165	1.3749	1.6416	3.5183	1.889177
Skewness	0.1884	-0.1123	0.2322	1.0788	0.317098
Minimum	289.0000	626.7512	6100.0000	381.0340	5555026
Maximum	1400.0000	1542.6210	8990.0000	1360.0000	14741260
Sum	20339.0000	28321.7500	192200.0000	16988.3500	2.47E+08
Count	26	26	26	26	26



Fig. 3. Growth rates of variables.

Table 3

correlation with field.				
Correlated variables	REC & GDPC	REC & NS	REC & UR	REC & CO2
correlation coefficients <i>t</i> -statistics P-value	0.961 -2.726 0.0489*	0.941 -3.467 0.0023*	0.982 4.529 0.0002*	0.772 -2.575 0.0177*

* denotes significance at the 5% threshold.

Table 4

Variables	REC	GDPC	CO2	NS	UR
REC	1.000				
GDPC	0.665*	1.000			
CO2	0.599*	0.633*	1.000		
NS	0.471*	0.928*	0.724*	1.000	
UR	0.478*	0.255*	0.686*	0.05*	1.000

* denotes correlations that are significant at 1% level.

(ADF) test, and the Phillip-Perron (PP) test [59]. The ADF tests, in fact, are based on the ordinary least squares estimation of the three model systems given by the relation Eq. (1).

$$\begin{aligned} \Delta y_t &= \rho y_{t-1} + \sum_{j=2}^p \boldsymbol{\Phi}_j \Delta y_{t-j+1} + \varepsilon_t \\ \Delta y_t &= \rho y_{t-1} + \sum_{j=2}^p \boldsymbol{\Phi}_j \Delta y_{t-j+1} + c + \varepsilon_t \\ \Delta y_t &= \rho y_{t-1} + \sum_{j=2}^p \boldsymbol{\Phi}_j \Delta y_{t-j+1} + bt + c + \varepsilon_t \end{aligned}$$
(1)

where t is the time index, c is an intercept constant called a drift, b is the coefficient on a time trend, ρ is the coefficient presenting process root, i.e. the focus of testing, p is the lag order of the first-differences autoregressive process, ϵ_t is an independent identically distributes residual term. The difference between the three equations concerns the presence of the deterministic elements c (a drift term) and bt (a linear time trend) [60].

These equations represent autoregressive models of order 1 with a constant and a trend, respectively. The specification error ϵ_t is assumed to be a process of white noise [61]. Machine execution results highlight a probability (P-value) at which the test performed is significant [58]. Generally, this probability is compared to the critical threshold (α) for rejecting or validating test results [17]. Thus, if $P - value < \alpha$ we accept the null hypothesis H_0 : there is at least one unit root. The process is said to be stationary at the critical threshold α . Otherwise, if $P - value > \alpha$, the null hypothesis is rejected : there is no unit root. The process is non-stationary. As a result, the stationarity of these chronological series must be determined using the sequence of first differences [62]. The results of the ADF tests are reinforced by the unit root tests developed by Philipps-Perron (PP) [63]. Indeed, ADF tests do not take into account any heteroscedasticity in the error term, but assume that the errors within the model are independent of each other and constitute white noise. The unit root test developed by Philipps and Perron (PP) overcomes the shortcomings of ADF tests and highlights the existence of a unit root more precisely. Constructed to take heteroskedastic errors into account, the test (PP) is subdivided into four steps structured as follows [64]:

- Ordinary least squares estimation of the three basic models of the Dickey–Fuller tests and calculation of the associated statistics and the residual e_t;
- Short term variance estimation: $\hat{\sigma}^2 = \frac{1}{n} \sum_{t=1}^n e_t^2$;
- Estimation of the corrective factor s_t^2 called long term variance, and defined by Eq. (2):

$$s_t^2 = \frac{1}{n} \sum_{t=1}^n e_t^2 + 2 \sum_{i=1}^l (1 - \frac{i}{l+1}) \left[\frac{1}{n} \sum_{t=i+1}^n e_t e_{t-i} \right]$$
(2)

where l is the number of lags defined by the Newey–West truncation as a function of the number of observations n: $l \approx 4(n/100)^{2/9}$; • Calculation of pp statistic by relation Eq. (3) :

$$pp = \sqrt{v} * \frac{(\hat{\phi} - 1)}{\hat{\sigma}_{\phi}} + \frac{n(v - 1)\hat{\sigma}_{\phi}}{\sqrt{v}}$$
(3)

with
$$v = \frac{\hat{\sigma}^2}{s^2}$$
 which takes the value 1 if e_t is white noise.

4.2. The optimal lag number

This refers to the time lag λ retained simultaneously by the majority of the chosen tests and capable of tracing the endogenous variable's trends as accurately as possible [65]. The Schwarz (Sc), Hannan-Quinn (HQ), Akaike (AIC) and Lagrande criteria are used to numerically select the right number of lags of the demand explanatory factors.

4.3. The Johansen cointegration test process

The Johansen test, is a procedure for testing cointegration of several, say k, I(1) time series. This test permits more than one cointegrating relationship [66]. There are two types of Johansen test, either with trace or with eigenvalue, and the inferences might be a little bit different. The null hypothesis for the trace test is that the number of cointegration vectors is $r = r^* < k$, vs. the alternative that r = k. Testing proceeds sequentially for $r^* = 1, 2, etc.$ and the first non-rejection of the null is taken as an estimate of r [66]. The null hypothesis for the maximum eigenvalue test is as for the trace test but the alternative is $r = r^* + 1$ and, again, testing proceeds sequentially for $r^* = 1, 2, etc.$, with the first non-rejection used as an estimator for r [66]. Once the stationarity of all variables in the same order has been established, the Johansen cointegration test is performed using trace statistics (λ_{trace}) and eigenvalue statistics (λ_{max}) [67]. At the 5% tolerance level, each of these statistics is compared to its number-based critical value. The possible orders of cointegration that may exist between the variables are tested according to the values of the number $r \in N$. The accepted order of cointegration is the one in which the values of (λ_{trace}) and (λ_{max}) are less than the critical values and significant at the 5% level [67].

4.4. ARDL approach for validating a long-run equilibrium

In order to validate the hypothesis of a long-run equilibrium between the variables, in addition to the Johansen test, this study uses ARDL modelling and upper-bound tests based on F-statistics. The ARDL technique has the advantage of being more efficient in the case of small and finite samples. In addition, the application of this technique makes it possible to obtain unbiased long-run estimates. According to Barkhordari et al. [68], the F-statistic is compared with the lower bound (Lb) and upper bound (Ub), respectively. If F-statistic > Ub, the hypothesis of non cointegration is rejected. If F-statistic < Lb, the null hypothesis of non cointegration cannot be rejected. Nevertheless, if Lb < F-statistic < Ub, the null hypothesis of no cointegration is not conclusive.

4.5. Vector error correction model

For a general VAR(p) model: There are two possible specifications for error correction: that is, two vector error correction models (VECM): the long run VECM and the transitory VECM [69]. The existence of cointegration means that there is at least one equilibrium relationship between the variables. According to Engel and Granger [70], the existence of an integral vector between the variables suggests that there is a causal relationship between them, at least in one direction. Since the variables are integrated we can continue the estimation of the error correction model that integrates short-term dynamics with long-term equilibrium [71,72].

The VECM makes it possible to detect the direction of causality in the long run, under the condition of stationarity of the different variables in first difference [73]. In this context, the error-correction term in the model must have a negative sign and be significant at the tolerance threshold [74]. Furthermore, the Wald causality test proves to be significant for assessing the direction of short-term causality, given the significance of the nullity test for the coefficients of the specified VECM model [75]. In the specific case where electricity consumption in the residential sector is the objective variable, the associated VECM is defined by the equation Eq. (4). In this equation, the white noise error term is ϵ_i , and the error correction terms is $ECT_{r,l-j}$ where *r* corresponds to the accepted order of cointegration as discussed in Section 4.3. ϕ_j denote the adjustment coefficients used to assess the level of imbalance correction within the model [59]. α is the intercept, *l*, *m*, *n*, *p* and *s* are lag numbers. β_i , γ_i , λ_i , ψ_i and η_i are estimated coefficients, Δ is the symbol for the first difference.

$$\Delta(REC)_{t} = \alpha + \sum_{i=1}^{l} \beta_{i} \Delta(REC)_{t-i} + \sum_{i=1}^{m} \gamma_{i} \Delta(GDPC)_{t-i} + \sum_{i=1}^{n} \lambda_{i} \Delta(CO_{2})_{t-i} + \sum_{i=1}^{p} \psi_{i} \Delta(NS)_{t-i} + \sum_{i=1}^{s} \eta_{i} \Delta(Ur)_{t-i} + \phi_{j} ECT_{r,t-j} + \varepsilon_{t}$$

$$(4)$$

4.6. Diagnostic tests

Diagnostic tests are used to validate the statistical consistency of the specified VECM and its relevance to the study of causal relationships. These include tests for Ramsey Reset specification, heteroskedasticity, normality of error terms and overall model stability [76–78].

4.6.1. Ramsey reset test

To ensure that the explanatory factors selected are sufficient to model residential electricity consumption, we carry out the Ramsey specification error test. The P values associated with the t-statistic, the F-statistic and the likelihood ratio of the test must be, respectively, greater than the tolerance threshold set at [79].

4.6.2. Heteroscedasticity test

When the variance of the error term within the model is not constant, the specified model is subject to heteroskedasticity [76]. This may be due to a number of statistical inconsistencies including:

- the repetition of the same value of the variable to be explained for different explanatory variable values;
- the relationship between errors and the values of an explanatory variable;
- The observations are averages calculated from various sample sizes.

Such circumstances may skew the quality of the results, resulting in incorrect analyses and conclusions. Several tests can be used to detect and correct heteroscedasticity in a time series. A Lagrange test, for example, can be carried out by computing the statistic $LM = n' * R^2$ (n'=n-p:difference between the number of observations n and the order of autocorrelation p) [77]. R^2 is the associated determination coefficient, which describes the averaging of the series' values. Then we compare $\chi^2_{\alpha}(p)$ to the α threshold in the order p. If $LM = n' * R^2 \sim \chi^2_{\alpha}(p)$, the hypothesis of error independence is rejected; in other words, the F-statistic and $\chi^2_{\alpha}(p)$ probabilities associated with the Lagrange test are less than the set critical value α . In this case, a first-order difference filter can correct the autocorrelation and produce a homoscedastic time series [77].

4.6.3. Normality test

The objective is to ensure that the ϵ_t error terms within the model are independent and identically distributed within the specified model, and therefore constitute white noise [78]. This test is based on the validity of the Jarque–Bera (JB) statistic defined by equation Eq. (5). The results of this test must be such that JB < 5.99 for a *P*-value > 5% [80], for the hypothesis of a normal distribution of the error terms within the model to be accepted.

$$JB = (n/6)\beta_1 + (n/24)(\beta_2 - 3)^2$$
(5)

 β_1 and β_2 are respectively coefficients of Skewness and Kurtosis.

Table 5 Outcomes of unit root test.

Variables	ADF test			PP-test	Decision		
	At level P-value	In first difference <i>P</i> -value	Lg	At level P-value	In first difference <i>P</i> -value	Lg	
REC	0.959	0.0001*	2	0.975	0.0001*	2	I(1)
GDPC	0.615	0.006*	2	0.569	0.005*	2	I(1)
CO2	0.171	0.034**	2	0.334	0.034**	2	I(1)
NS	0.875	0.0127**	2	0.875	0.013**	2	I(1)
UR	1.000	0.0015*	2	1.000	0.0000*	2	I(1)

(*), (**) indicate significance at 1% and 5% respectively.

Lg indicates the Lag length of the dependent variable.

The Lag length for the ADF was selected using Schwarz information Criterion.

Bandwidth: 2 (Newey-West automatic) using Bartlett kernel.

4.6.4. Model stability test

The stability test is performed using the cumulative sum of the recursive residuals S(t) and the cumulative sum of the squares of the recursive residuals S'(t) defined by the Eqs. (6) and (7) suggested by Brown et al. [81]. The trends in the S(t) and S'(t) statistics must be contained within the interval defined by the relationship Eq. (8) for the validation of the long run stability hypothesis of the model specified [82]. Furthermore, as a last step to analyse the stability of the ARDL model estimated in this study, we check all its inverse roots with respect to the unit cycle. All the roots must be contained in the unit circle to validate the stability of the model.

$$S(t) = (n-k) \left(\sum_{j=k+1}^{t} \tilde{\varepsilon}_j \right) / \left(\sum_{j=k+1}^{t} \tilde{\varepsilon}_j^2 \right)$$
(6)

$$S'(t) = \left(\sum_{j=k+1}^{r} \tilde{\varepsilon}_{j}^{2}\right) / \left(\sum_{j=k+1}^{r} \tilde{\varepsilon}_{j}^{2}\right)$$
(7)

 $(t = k+1, \dots, n)$. The number of variables is k, and the number of observations is n.

$$I = \left[\pm\beta \left(2t + n - 3k\right)/\sqrt{(n-k)}\right]$$
(8)

In this relation $\beta = 0.948$ at the critical threshold $\alpha = 5\%$ [83].

5. Empirical results

This section presents the results of statistical tests relating to the modelling of electricity consumption as a function of explanatory factors. Overall, the statistical analysis highlights the existence of unidirectional causality from all the explanatory factors to electricity consumption in the short and long runs, respectively. Therefore, reliable development strategies can be formulated for decision-makers to help manage electricity demand in Cameroon's residential sector.

5.1. Stationarity assessments

Table 5 shows the results of the stationarity test for the different variables. At level, the ADF and PP tests have P-values above the critical threshold of 5%. For first difference variables, the P-values associated with the tests are well below the 5% threshold. As mentioned in Section 4.1, the variables used in this analysis are non-stationary at level and stationary at first difference. This result suggests the existence of long run cointegration between the variables [84]. However, the cointegration test should fully confirm this observation.

5.2. Optimal lag number

According to the Schwarz (SC), Hannan-Quinn (HQ), Akaike (AIC), Final prediction error (FPE) and Lagrange (LR) criteria, l = m = n = p = s = 2 is the best lag number as shown in Table 6. As specified in Section 4.2, the time response function is therefore a linear twoperiod combination of electricity consumption (ΔREC)_t and the set of explanatory factors.

adle 6	
elected	ontima

Selected optimal lag number.						
Lag	LogL	LR	FPE	AIC	SC	HQ
0	-983.01	NA	3.94e+29	82.33	82.57	82.39
1	-788.11	292.33	2.95e+23	68.17	69.64	68.56
2	-725.54	67.78*	1.77e+22*	65.04*	67.74*	65.76*

* Indicates the lag order currently selected by the criteria.

5.3. Outcomes of Johansen cointegration test

Table 7 presents the results of an incremented test of the order of cointegration of the various variables based on the estimators (λ_i , race) and (λ_{\max}) . We see that for a null order of cointegration, r = 0, $\lambda_t race =$ 131.239, and $\lambda_m ax = 68.876$ are greater than their respective critical values of 69.818 and 33.876. The null hypothesis of cointegration is rejected. The previous observation holds true for r = 1 and r = 2. The cointegration null hypothesis is rejected once more because the estimators remain above their critical values. We see that for r = 3, λ_{trace} = 9.026 < 15.494 and λ_{max} = 6.767 < 14.264. At the tolerance threshold of 5%, the hypothesis of three cointegrating relationships between the five correlated variables is accepted. Thus, the order of cointegration highlighted by this test is r = 3 as outlined in Section 4.3. Each estimated model therefore contains three error-correction terms. The hypothesis of a long run equilibrium between the variables is accepted. However, as the data used in this analysis cover only 26 observations, we also perform upper bound tests to verify the results of the Johansen tests, in relation to the approach proposed in Section 4.4.

5.4. ARDL modelling and long run equilibrium of variables

5.4.1. ARDL modelling and statistical proofing

Before evaluating the long run cointegration by the ARDL technique, it is necessary to check the statistical validity of the model. Table 8 shows the statistics related to the estimation of the ARDL model, with electricity consumption as the dependent variable. The R^2 , the adjusted R^2 , the P-values, and the probability of the F-statistic are better for a maximum number of lags mostly equal to 2. Exception observed for GDP per capita (GDPC), where the optimal number of lags is 1. This difference supports the hypothesis of an almost immediate effect on the REC variable of any shock observed within the GDPC.

5.4.2. Serial correlation and heteroskedasticity tests for ARDL model

The results of the Breusch–Godfrey serial correlation LM test presented in Table 9 show that the *P*-value associated with the test is 16.75% > 5%. The hypothesis of the existence of a serial correlation between the variables is rejected at the 5% threshold. Moreover, the *P*value associated with the $\chi^{(2)}$ of the heteroskedasticity test is 51.45% > 5%. As indicated in Section 4.6.2, the residuals of the model are homoscedastic, identically and independently distributed. They therefore constitute white noise.

Table 8

Dutcomes of Johansen cointegration tests.							
Rank (r)	λ_{trace}	P-value***	λ_{trace} critical value	$\lambda_{ m max}$	P-value***	λ_{\max} critical value	
r=0	131.2391	0.000*	69.8188	68.8760	0.000*	33.8768	
r=1	62.3630	0.0012*	47.8561	30.7440	0.0190*	27.5843	
r=2	31.6190	0.0305*	29.7971	22.5923	0.0309*	21.1316	
r=3	9.0266	0.3629**	15.4947	6.7675	0.5170**	14.2646	

 * denotes rejection of the nomber of cointegrating equations at 5% level.

** denotes acceptation of the nomber of cointegrating equations at 5% level.

*** MacKinnon-Haug-Michelis (1999) p-values, at the 5% critical threshold.

ARDL modelling.				
Variable	Coefficient	Std.Error	t-statistic	P-value
С	-801.31	314.05	-2.55	0.03
$\Delta(REC(-1))$	0.39	0.31	1.27	0.23
$\Delta(REC(-2))$	0.51	0.25	1.96	0.08
$\Delta(GDPC(-1))$	-0.44	0.15	-2.85	0.02
$\Delta(CO2(-1))$	-0.02	0.03	-0.69	0.50
$\Delta(CO2(-2))$	0.07	0.02	3.44	0.008
$\Delta(NS(-1))$	3.33	0.90	3.68	0.006
$\Delta(NS(-2))$	2.25	0.65	3.44	0.008
$\Delta(UR(-1))$	-0.06	0.01	-4.30	0.002
$\Delta(UR(-2))$	0.06	0.01	4.19	0.003
REC(-1)	-1.31	0.36	-3.64	0.006
GDPC(-1)	-0.08	0.20	-0.41	0.69
CO2(-1)	9.44E-05	0.02	0.003	0.99
NS(-1)	-4.20	1.08	-3.88	0.004
UR(-1)	0.0005	0.0002	2.19	0.059
R^2	0.912			
Adjusted R ²	0.76			
Log Likelihood	-100.93			
F-statistic	5.97			
Prob.(F-statistic)	0.007			
ECT	-1.838			

Dependent variable: $\Delta(REC)$.

Method: least squares.

Sample (adjusted): 1997-2019.

* indicates significance at 5% threshold.

Table 9

LM and Heteroscedasticity tests for ARDL model.								
Serial correlation LM test	F-statistic	obs*R-squared	Prob.F	Prob.chi2				
	0.551	-	0.602	0.167				
Heteroscedasticity test	0 763	13 15	0.685	0 514				

Wald test for ARDL model coefficients.							
Statistic	Value	df	Probability				
F-statistic	6.65	(7, 8)	0.0079				
chi2	46.60	7	0				
-							

Null hypothesis: C(4) = C(5) = ... = C(10) = 0.

5.4.3. Cusum tests results for ARDL model

Fig. 4 shows the Cusum and Cusum of squares of the model contained within the 95% confidence interval. The specified ARDL model is therefore statistically robust. This model can be used to perform a long run cointegration analysis between the variables.

5.4.4. Upper bounds tests

By assigning the explanatory factors $\Delta(GDPC(-1))$, $\Delta(CO2(-1))$, $\Delta(CO2(-2))$, $\Delta(NS(-1))$, $\Delta(NS(-2))$, $\Delta(UR(-1))$ and $\Delta(UR(-2))$ the notations C(4), C(5), C(6), C(7), C(8), C(9) and C(10) in accordance with the order of appearance in the model, the non-cointegration of the variables is consistent with hypothesis H0: C(4) = C(5) = C(6) = C(7) = C(8) = C(9) = C(10) = 0, while the alternative hypothesis is H1: $C(4) \neq C(5) \neq C(6) \neq C(7) \neq C(8) \neq C(9) \neq C(10) \neq 0$. Table 10 shows the results of the Wald test. The F-statistic associated with the test is F = 6.65 for an associated *P*-value of 0.79%. This value is compared to the critical value of Pesaran et al. [68], at 5%. Our model being unrestricted with an intercept and no trend, the upper bound corresponding to the F-statistic = 6.65 is Ub = 4.85, while the lower bound is Lb = 3.78. We see that F-statistic = 6.65 > 4.85.

alternative hypothesis H1 is accepted. There is a long run cointegration relationship between the variables as mentioned in Section 4.4. The estimation of the error correction term indicated in Table 8 gives ECT(-1) = -1.838. The system is getting toward long run equilibrium at a speed of 183.8%.

5.5. Vector error correction models estimation

Table 10

Table 11 presents the estimated coefficients of the model specified by Eq. (4) for $\Delta(REC)_t$ as objective function. By integrating the coefficients values estimated by the software, the objective function $\Delta(REC)_t$ is rewritten in the form given by Eq. (9).

$$\begin{split} & \varDelta(REC)_t = -1.629((REC)_{t-1} - 0.04(CO_2)_{t-1} + 0.00015(UR)_{t-1} - 1996.529) \\ & -4.731((NS)_{t-1} + 0.0076(CO_2)_{t-1} - 0.0002(UR)_{t-1} + 1359.98) \\ & -0.235((GDPC)_{t-1} - 0.046(CO_2)_{t-1} + 0.0005(UR)_{t-1} - 6340.36) \\ & +0.648\varDelta(REC)_{t-1} + 0.720\varDelta(REC)_{t-2} + 3.99\varDelta(NS)_{t-1} + 2.63\varDelta(NS)_{t-2} \\ & -0.69\varDelta(GDPC)_{t-1} - 0.18\varDelta(GDPC)_{t-2} - 0.06\varDelta(CO_2)_{t-1} + 0.06\varDelta(CO_2)_{t-2} \\ & -0.06\varDelta(UR)_{t-1} + 0.05\varDelta(UR)_{t-2} + 3797.96 + \varepsilon_t \end{split}$$

Functional	Model	Coefficients	P-value	Coefficients	P-value	ECT	P-value
variable	description	at j = 1	at j = 1	at j = 2	at j = 2		
	$\Delta(REC)$.	0.648	0.015	0.720	0.002	-1.629	0.0005*
	$\Delta(GDPC)_{i=j}$	-0.691	0.003	-0.187	0.190	-0.235	0.0034*
$\Delta(REC)$	$\Delta(CO2)$	-0.062	0.011	0.065	0.007	-4.731	0.0007*
()/	$\Delta(NS){i=j}$	3.999	0.0008	2.635	0.0017		
	$\Delta(UR)_{i=j}$	-0.066	0.0006	0.058	0.0011		
	_(* *)/=j						
R^2	0.905	Adjusted R ²	0.769	F-stat.	6.641	Prob.(F-stat.)	0.003
	$\Delta(GDPC)_{t-j}$	0.559	0.084	0.267	0.249	-0.107	0.835
	$\Delta(REC)_{t-j}$	0.442	0.245	0.352	0.256	-1.379	0.651
	$\Delta(CO2)_{t-j}$	0.004	0.888	0.001	0.964	-2.661	0.118
$\Delta(GDPC)_{t}$	$\Delta(NS)_{t-j}$	2.238	0.129	-0.244	0.808		
	$\Delta(UR)_{t-j}$	-0.036	0.126	0.054	0.023		
- 2				_			
R^2	0.863	Adjusted R ²	0.665	F-stat.	4.372	Prob.(F-stat.)	0.016
	$\Delta(NS)_{t-j}$	0.091	0.951	-0.016	0.988	0.012	0.983
	$\Delta(REC)_{t-j}$	0.035	0.930	-0.086	0.793	0.243	0.586
	$\Delta(GDPC)_{t-j}$	-0.086	0.792	0.031	0.898	-1.663	0.353
$\Delta(NS)_t$	$\Delta(CO2)_{t-j}$	0.035	0.342	-0.0218	0.542		
	$\Delta(UR)_{t-j}$	-0.028	0.269	0.017	0.451		
P ²	0 733	Adjusted R^2	0 347	F-stat	1 902	Prob (E-stat)	0 168
K	0.755	Aujusteu A	0.347	1-3tat.	1.902	1100.(1-3tat.)	0.100
		0.000	0.005	0.050	0.040	5.0.40	0.000
	$\Delta(CO2)_{t-j}$	0.022	0.935	0.263	0.342	5.342	0.236
	$\Delta(REC)_{t-j}$	0.400	0.896	0.134	0.756	-1.954	0.566
1(000)	$\Delta(GDPC)_{l-j}$	-1.125	0.052	3.12/	0.120	-24.239	0.093
$\Delta(CO2)_{l}$	$\Delta(NS)_{t-j}$	24.2/3	0.059	12.1//	0.173		
	$\Delta(UR)_{t-j}$	-0.2/8	0.156	0.178	0.318		
R ²	0.635	Adjusted R^2	0.108	F-stat	1 205	Prob (F-stat)	0 398
R	0.000	Rujusteu R	0.100	i stat.	1.200	1100.(1 5.00.)	0.000
	$\Lambda(IIR)$	1 620	0.0001	_0 789	0.0048	_1 781	0 743
	$\Delta(O R)_{t-j}$	1.047	0.734	2 045	0.350	2 652	0.208
	A(GDPC)	-5 449	0.106	2.743 _3.201	0.339	_25 201	0.152
$\Lambda(\mathbf{I}, \mathbf{P})$	$\Delta(ODTC)_{l=j}$	0.246	0.100	-3.221	0.195	-23.271	0.132
$\Delta(O R)_{l}$	$\Delta(O_2)_{l=j}$	1 620	0.0001	0.701	0.195		
	$\Delta(1 \vee D)_{l-j}$	1.047	0.0001	-0.707	0.00-0		
R^2	0.999	Adjusted R ²	0.999	F-stat.	43,435,98	Prob.(F-stat.)	0.000
		· j			,		

(*) indicates significance at 1% threshold.

(9)

5.5.1. VECM and assessment of long run causality

For the four explanatory factors, namely, $\Delta(GDPC)_t$, $\Delta(NS)_t$, $\Delta(CO2)_t$, and $\Delta(UR)_t$ the estimated coefficients are also presented. The estimated model for these four explanatory factors are derived in a similar way as Eq. (9) in order to provide a more detailed analysis of causality. So we have five equations based on the change in the functional variable among the five variables chosen.

For the model with the functional variable $\Delta(REC)_t$, the three associated error correction terms are negative and significant at the tolerance threshold of 1%. This finding supports the hypothesis of longrun causality from all explanatory factors considered simultaneously to electricity consumption. We can make a definitive statement about the direction of the interaction by observing the signs and significance of the error-correction terms in the other four models. Consistent with the results presented in Table 11, the error correction terms in the functional variable models $\Delta(GDPC)_t$, $\Delta(NS)_t$, $\Delta(CO2)_t$, and $\Delta(UR)_t$ are insignificant at the 1% tolerance level. According to the $\Delta(REC)_t$ model, long run causality is thus unidirectional, from all explanatory factors to electricity consumption.

5.5.2. Wald tests and assessment of short-term causality

An analysis of the direction of short-term causality is performed using the model in Table 11. The Wald causality test results are presented in Table 12. The tests on the nullity of the directing coefficients C(i) associated with each explanatory factor are significant at the 1% threshold for the functional variable model $\Delta(REC)_i$. This finding supports the hypothesis of short-term causality from each explanatory factor to electricity consumption. The observations of the models $\Delta(GDPC)t$, $\Delta(GDPC)t$, $\Delta(CO2)t$, $\Delta(NS)t$, and $\Delta(UR)t$ reveal that the results of the tests on the nullity of the coefficients are globally non-significant. The results in bold indicate that there is no short-term causality from electricity consumption to each of the explanatory factors. As a result, the short-term causal relationship discovered is unidirectional, running from all explanatory factors to electricity consumption. Additional statistical tests are used to validate the effectiveness of the identified causal relationship.

5.6. Diagnostic tests

5.6.1. Ramsey reset test results

Table 13 shows the results of the regression error specification test, performed with a number of fitted terms equal to 1. The P-values of the t-statistic, F-statistic, and Likelihood ratio are all above the tolerance threshold of 5%. The explanatory factor $\Delta(GDPC)$, $\Delta(CO2)$, $\Delta(NS)$, and $\Delta(UR)$ are considered necessary and sufficient to explain the trends of the dependent variable $\Delta(REC)$. Any modelling based on these variables is free of specification errors.

Explained Explanatory N		Null	$\chi^{(2)}$	df	P-value
variable	variables	hypothesis			
	$\Delta(REC)_{t-j}$	C(4) = C(5) = 0	16.995	2	0.0002*
	$\Delta(GDPC)_{t-j}$	C(6) = C(7) = 0	17.960	2	0.0001*
$\Delta(REC)_{t}$	$\Delta(NS)_{t-j}$	C(8) = C(9) = 0	16.909	2	0.0002*
	$\Delta(CO2)_{t-j}$	C(10) = C(11) = 0	25.240	2	0.0000*
	$\Delta(UR)_{t-j}$	C(12) = C(13) = 0	26.075	2	0.0030*
		Q(10) Q(10) 0	1.064		
	$\Delta(REC)_{t-j}$	C(18) = C(19) = 0	1.864	2	0.393
	$\Delta(GDPC)_{t-j}$	C(20) = C(21) = 0	3.767	2	0.152
$\Delta(GDPC)_t$	$\Delta(NS)_{t-j}$	C(22) = C(23) = 0	0.028	2	0.985
	$\Delta(CO2)_{t-j}$	C(24) = C(25) = 0	11.393	2	0.0034
	$\Delta(UR)_{t-j}$	C(26) = C(27) = 0	11.685	2	0.0029
	A(REC)	C(32) = C(33) = 0	0.109	2	0.946
	$\Delta(GDPC)$	C(34) = C(35) = 0	7 407	2	0.024
$\Delta(CO2)$.	$\Delta(NS)$	C(36) = C(37) = 0	1.141	2	0.565
_(==)/	$\Delta(CO2)$	C(38) = C(39) = 0	4.986	2	0.082
	$\Delta(UR)_{t-j}$	C(40) = C(41) = 0	3.235	2	0.198
	$\Delta(REC)_{t-j}$	C(46) = C(47) = 0	0.179	2	0.914
	$\Delta(GDPC)_{t-j}$	C(48) = C(49) = 0	0.243	2	0.885
$\Delta(NS)_t$	$\Delta(NS)_{t-j}$	C(50) = C(51) = 0	1.143	2	0.564
	$\Delta(CO2)_{t-j}$	C(52) = C(53) = 0	0.018	2	0.990
	$\Delta(UR)_{t-j}$	C(54) = C(55) = 0	1.945	2	0.378
			1		
	$\Delta(REC)_{t-j}$	C(60) = C(61) = 0	1.030	2	0.59/
((TT B)	$\Delta(GDPC)_{t-j}$	C(62) = C(63) = 0	3.313	2	0.190
$\Delta(UR)_t$	$\Delta(NS)_{t-j}$	C(64) = C(65) = 0	2.081	2	0.353
	$\Delta(CO2)_{t-j}$	C(66) = C(67) = 0	2.214	2	0.330
	$\Delta(UR)_{t=i}$	C(68) = C(69) = 0	105.310	2	0.000

j = (1; 2) is the time lag of the explanatory factors, and for which the coefficients C(i) are tested. (*) denotes the test's significance at 1% threshold.

Table 13

Ramsey reset test findings.					
Statistics	Value	df	P-value		
t-statistic	0.381	19	0.707		
F-statistic	0.145	(1, 19)	0.707		
Likelihood ratio	0.190	1	0.663		

Table 14

Heteroscedasticity tests.					
Test	est Statistics P-value		Observation		
	$\chi_p^{(2)} = 0.083$	0.920	P-value > 5%		
LM-test			Homoscedasticity		
	F-statistic = 0.783	0.548	of error terms		

5.6.2. Heteroscedasticity test for VECM

As seen in Table 14, the probabilities associated respectively with the F-statistic and the $\chi_p^{(2)}$ are relatively higher than the error threshold of 5%. Consequently, the model is homoscedastic at the critical threshold of 5%. There is no correlation between variations in GDPC, CO2, NS and UR, but there is a direct correlation between each of these explanatory factors and residential electricity consumption.

5.6.3. Normality test

The results of this test confirm the earlier conclusion. With a confidence level of 95%, the distribution of the error terms in the $\Delta(REC)_r$ model follows a Gaussian distribution. Consequently, the residuals of the model are independently and identically distributed (iid). Any statistically valid explanatory factor added to this model will have no effect on the four existing explanatory factors, but a direct correlation with electricity consumption.

5.6.4. Stability tests

Cusum and Cusum-squared curves are contained within the range of 95%, as shown in Fig. 4. The $\Delta(REC)t$ model has been shown to be statistically valid and stable. Since there is unidirectional causality from all the explanatory factors to residential electricity consumption, any impetus within one of these factors will have a direct effect on consumption, without feedback.

As a final step to analyse the stability of the estimated VECM, we check all its inverse roots in relation to the unit cycle. Fig. 5 presents the results of the inverse roots of AR characteristic polynomial. All the roots are inside the unit cycle, confirming the stability of the model.

6. Discussions

In this section, we discuss how the explanatory factors selected can contribute to ensuring the sustainability of electricity consumption in the residential sector over the short and long run. This is in line with the results highlighted in Section 5. Strategies for better managing electricity consumption are also outlined.

6.1. GDP per capita as sustainability factor of residential electricity consumption

According to [20], GDP had a significant impact on residential electricity consumption in Algeria from 1970 to 2013. Any increase in Algeria's GDP leads to an increase in residential electricity consumption. Our findings support this hypothesis. With a unidirectional causality from GDP per capita to electricity consumption (see Sections 5.5.1 and 5.5.2), any increase in GDP per capita in Cameroon would result in an increase in electricity consumption. Improving per capita GDP is a pre-requisite for achieving the 7-1 Sustainable Development Goals. With a GDP per capita of around USD 1500 in 2019 [8], just under USD 5 per day, it is difficult for a cameroonian household to meet its basic needs and subscribe to a regular energy service. This causes problems like less access to energy services, more unpaid bills, and the spread of electricity fraud because of uncontrolled connections that the national electricity distributor doesn't know about. Per capita GDP thus appears to be a key factor in the sustainability of electricity consumption in the residential sector. This factor is closely linked to the proliferation of electrical appliances in households, with the corollary of increased electricity consumption in the residential sector [85].

6.2. Compact urbanisation for long run sustainability of residential electricity consumption

Urbanisation remains inextricably linked to electricity consumption in the Cameroonian residential sector, just as it was in Tunisia in 2020, according to [21]. However, the opposite directions of causality show that population migration from rural to urban areas increases electricity demand. Cameroon's urbanisation rate is increasing. According to [8], 58.15% of the population will live in cities by 2021, up from 55.18% in 2016. In this situation, it is critical for the government to design an urban model that allows for the containment of displaced populations, the implementation of energy efficiency measures, and the control of electricity demand in order to satisfy it. The urbanisation in Cameroon follows mostly the spread-out urbanisation model as opposed to the compact urbanisation model. Cities grow under this model by consuming more land and air space. As a result, the overhead electricity network is more branched and extended over long distances. As a result, power poles are spread across a larger land area. This increases the likelihood of incidents on electricity networks as well as the projected maintenance costs. Limiting the spread of construction is therefore necessary to combat this type of expensive and energyconsuming urban planning. Sustainable urban planning, which includes limiting urban sprawl, would help Cameroon's electricity system and



Fig. 4. Cusum and Cusum of squares tests.

residential sector develop more sustainably. The adoption of the compact urban model, supported by energy efficiency measures shared by all residences, will ensure the long-term sustainability of the electricity demand imposed by urbanisation.

6.3. CO2 emissions and residential electricity consumption

Table 15 depicts additional factors that influence residential electricity consumption. The role of CO2 in influencing residential electricity demand is not by chance. In fact, hydroelectric production is still insufficient to meet the growing population's electricity demands. As a solution to this problem, Cameroon's government has chosen to generate electricity through oil or gas power thermal stations. The release of CO2 into the atmosphere is a direct result of these conditions. In fact, Cameroon has approximately twenty thermal power stations that run on heavy or light fuel oil, with a total capacity of 374.26 MW in 2015. There are also two operational gas power plants with a combined capacity of 266 MW. The combined installed capacity of the three operational hydroelectric power stations is 732.2 MW. In 2015, heavy fuel oil, light fuel oil, and gas-fired thermal power stations accounted for 46.65% of the total installed production capacity of 1372.46 MW. Furthermore, self-generators, which accounted for 22% of total production capacity in 2016 [9], run primarily on diesel, contributing to CO2 emissions from self-generation of electricity. Furthermore, the residential sector's estimated electricity consumption is 1383 GWh, which is lower than the industrial sector's estimated consumption of 3619 GWh in 2019 [1]. This value order indicates that a significant portion of thermal production is made available to households, SMIs, and SMEs.

is such that generation follows demand. The electricity produced by thermal power stations in Cameroon is used to support hydroelectric generation in order to meet demand from the residential sector. Consequently, any increase in thermal generation leads to an increase in CO2 emissions and a satisfaction of household electricity consumption. In other words, the increase in CO2 emissions from thermal electricity generation helps to meet the demand for electricity from the residential sector, and therefore has a positive effect on consumption. However, the opposite effect is not necessarily true. An increase in electricity consumption in the residential sector may not lead to an increase in CO2 emissions. In fact, the increase in GDP per capita, the proliferation of energy-consuming household appliances, the increase in the number of electricity consumers per household, the proliferation of small and medium-sized industries and small and medium-sized enterprises that use electricity to produce goods and services, can all support an increase in residential electricity consumption, without any increase in CO2 emissions due to thermal electricity generation.

The electricity supply system for the residential sector in Cameroon

Taking into account that CO2 emissions due to hydroelectricity are negligible [86], this analysis remains consistent with the unidirectional causality from CO2 emissions to residential electricity consumption as outlined in Sections 5.5.1 and 5.5.2. This means that thermal electricity generation and its associated CO2 emissions are central to the sustainability of residential electricity consumption in Cameroon. Cameroon still needs to step up its thermal generation in order to sustainably support residential demand. However, this intensification must be accompanied by a plan for the gradual reduction of CO2 emissions in order to contribute effectively to the objective of reducing national emissions by 35% by 2030, as agreed by the State of Cameroon

Table	e 15	

Author's contribution.						
Sector	Authors	Country	Year of study	Empirical method	Period	Results
	[20]	Algeria	2018	ARDL	1970–2013	$GDP \rightarrow REC$
Residential	[21]	Tunisia	2020	Granger Causality	1980–2018	$\begin{array}{l} \text{REC} \rightarrow \text{EP} \\ \text{REC} \rightarrow \text{UR} \end{array}$
	Writers	Cameroon	2022	ARDL, VECM	1994–2019	$GDPC \rightarrow REC$ $CO2 \rightarrow REC$ $UR \rightarrow REC$ $NS \rightarrow REC$

Note : \rightarrow represents unidirectional.



Fig. 5. Inverse roots of AR characteristic polynomial.

in 2021 at the 26th conference of the parties (COP26) in Glasgow, United Kingdom. This objective of -35% of CO2 emissions can be achieved by significantly involving the residential sector. This will involve stepping up hydroelectric, solar and wind generation, which account respectively for 66.48%, 0.24% and 0.0% of the country's electricity generation in 2020 [1]. CO2 emissions therefore appear to be a major energy factor in the sustainability of electricity demand in the residential sector.

6.4. The number of subscribers and the sustainability of residential electricity consumption

An increase in the number of subscribers to the distribution network leads directly to an increase in electricity consumption. This logical consequence supports the idea of a unidirectional causality from the number of subscribers to residential electricity consumption. However, the increase in the number of subscribers to the electricity network highlights two problems: saturation of the network and deterioration in the quality of supply, with the corollary of unsatisfied demand. To ensure that all Cameroonians have access to reliable and modern electricity services, the government should combat fraudulent connections to the distribution networks, renovate obsolete and saturated power lines, and carry out extension work. The number of subscribers, which is often neglected in causality studies, is therefore a determining factor in the long run sustainability of electricity demand in the residential sector.

7. Conclusions and policy implications

The analysis of causal relationships involving various factors of electricity demand sustainability is a useful approach for identifying energy strategies to be implemented in order to contribute to the long-term development of various consumption sectors. The residential sector is one of the energy consumption sectors in Cameroon that has seen an increase in demand for electricity over the last ten years. The scarcity of literature in Cameroon on studies of the causal relationship between electricity demand and its explanatory factors justifies the importance of this reflection. In fact, between 2012 and 2022, only two scientific studies in the African region looked closely at the possible causal links between residential electricity consumption and its main factors. This also explains why there are not many energy strategies for decision-makers that could help the residential sector reach its goals for sustainable development.

Controlling variations in the most influential factors that dictate the direction of electricity fluctuations on transmission and distribution networks is critical to ensuring universal access to adequate electricity. In addition, to GDP and urbanisation, which have been identified as factors with a direct causal relationship with residential electricity consumption in Africa in the literature, this study identifies GDP per capita, CO2 emissions from thermal electricity generation, and the evolution of the number of subscribers, as additional parameters to be included in the analysis of causal relationships involving residential electricity consumption. The Johansen cointegration model, the autoregressive distributed lag model, the vector error correction model, and the Wald-Granger causality tests are still very useful for analysing cointegration between variables and determining causality direction over time.

In both the short and long run, the results show unidirectional causality from each explanatory factors to residential electricity consumption. An increase in GDP per capita has been identified as a necessary measure to improve consumers' ability to afford a regular energy service. The spread-out model of urbanisation currently being observed in Cameroon favours a strong branching of distribution lines, uncontrolled electrical interconnections, and increased fraud. In this situation, it is recommended that a compact urban model be adopted. Implementing such a measure in the majority of urban areas will limit power line overhead densification, reduce the risk of impacts, impose common energy efficiency measures on residences, and better control electricity consumption. The study's recommendations include reducing CO2 emissions from thermal electricity generation and further controlling the evolution of the number of subscribers on distribution networks, which will aid in the achievement of sustainable development goals in the residential sector. CO2 capture and storage is one of the technologies that will be used to reduce the carbon footprint of thermal power plants. Prepaid metres can be used to ensure that electricity consumption is proportional to each subscriber's wallet. This would prevent unpaid bills from causing power outages and disputes between the electricity distributor and its customers.

Although statistical tests indicate that the ARDL and VECM models in this study are stable and free of specification errors, other variables such as the price of electricity, the number of households, thermal electricity generation, hydroelectric generation, and per capita electricity consumption, can be included in the models in order to shed light on their respective effects on electricity demand, and to bring out more factors in the sustainability of electricity consumption in the residential sector.

CRediT authorship contribution statement

Serge Guefano: Conceptualization, Data curation, Software, Methodology, Investigation, Preparation, Writing – original draft, Writing – review & editing. **Mokhtar Bozorg:** Formal analysis, Methodology, Writing – original draft, Writing – review & editing, Validation, Funding acquisition, Resources. **Mauro Carpita:** Supervision, Writing – original draft, Project administration, Funding acquisition, Resources, Validation. **Jean Gaston Tamba:** Supervision, Conceptualization, Formal analysis. **Jacques Etame:** Supervision, Funding acquisition.

Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Data availability

I give permission for my data to be published at the same time as this work. I make my data available in this submission

Acknowledgments

The Swiss Confederation's grant reference 2022/0694/kamerun.op for international students is funding this project. We would like to thank the Swiss Confederation, which spares no effort to support the research work of young scientists from developing countries.

Appendix A. Supplementary data

Supplementary material related to this article can be found online at https://doi.org/10.1016/j.esr.2023.101155.

References

- [1] International Energy Agency data base, 2022, https://www.iea.org.
- [2] Jean Gaston Tamba, Flavian Emmanuel Sapnken, Tchitchile Wilfried Emmanuel Azong, et al., An overview of electricity in Cameroon: current status, influential factors and government actions, Int. J. Energy Econ. Policy 12 (4) (2022) 470–481.
- [3] Jean Calvin Nsangou, Joseph Kenfack, Urbain Nzotcha, et al., Assessment of the potential for electricity savings in households in Cameroon: A stochastic frontier approach, Energy 211 (2020) 118576, http://dx.doi.org/10.1016/j.energy.2020. 118576.
- [4] Fernand Ghokeng, Cameroon: The Government Raises the Cost of Electricity for Companies, Sikafinance, 2023, https:// www.sikafinance.com.
- [5] Jithesh Arayankalam, Satish Krishnan, ICT-based country-level determinants of social media diffusion, Inf. Syst. Front. (2022) 1–22.
- [6] Azangue Willy, Deciphering Medium-Voltage Energy Billing in Cameroon, LinkedIn, 2023, https://www.fr.LinkedIn.com.
- [7] Mohammed Yekini Suberu, Mohd Wazir Mustafa, Nouruddeen Bashir, et al., Power sector renewable energy integration for expanding access to electricity in sub-Saharan Africa, Renew. Sustain. Energy Rev. 25 (2013) 630–642, http: //dx.doi.org/10.1016/j.rser.2013.04.033.
- [8] World bank data base, 2022, https://data.worldbank.org.
- [9] Ministry of water ressources and energy, Energy balance sheet of Cameroon, 2016, https://www.mng-cameroon.org.
- [10] Nzoko Tayo Dieudonné, Talla Konchou Franck Armel, Aloyem Kaze Claude Vidal, et al., Prediction of electrical energy consumption in Cameroon through econometric models, Electr. Power Syst. Res. 210 (2022) 108102, http://dx.doi. org/10.1016/j.epsr.2022.108102.
- [11] Pierre-Olivier Pineau, How sustainable is policy incoherence? A rationalist policy analysis of the Cameroonian electricity reform, J. Clean. Prod. 15 (2) (2007) 166–177, http://dx.doi.org/10.1016/j.jclepro.2005.09.002.
- [12] Laura Scherer, Paul Behrens, Arjan De Koning, et al., Trade-offs between social and environmental sustainable development goals, Environ. Sci. Policy 90 (2018) 65–72, http://dx.doi.org/10.1016/j.envsci.2018.10.002.

- [13] Remember Samu, Festus Victor Bekun, Murat Fahrioglu, Electricity consumption and economic growth nexus in Zimbabwe revisited: fresh evidence from Maki cointegration, Int. J. Green Energy 16 (7) (2019) 540–550, http://dx.doi.org/10. 1080/15435075.2019.1598417.
- [14] Burcu Ozcan, Ilhan Ozturk, Renewable energy consumption-economic growth nexus in emerging countries: A bootstrap panel causality test, Renew. Sustain. Energy Rev. 104 (2019) 30–37, http://dx.doi.org/10.1016/j.rser.2019.01.020.
- [15] A. Matthew, Electricity consumption, government expenditure and sustainable development in Nigeria: A co-integration approach, 2019, 670216917, http: //dx.doi.org/10.32479/ijeep.7547.
- [16] Akinwale Yusuf Opeyemi, Muzindutsi Paul-Francois, Electricity consumption, trade openness and economic growth in South Africa: An ARDL approach, J. Econ. Coop. Dev. 40 (1) (2019) 135–156, https://www.researchgate.net/profile/ Yusuf-Akinwale/publication/334737621.
- [17] Mbanda L. Njoke, Empirical analysis of electricity consumption, CO2 emissions and economic growth: Evidence from Cameroon, 2019, 670216917, http://dx. doi.org/10.32479/ijeep.7915.
- [18] Hamisu Sadi Ali, Solomon Prince Nathaniel, Gizem Uzuner, et al., Trivariate modelling of the nexus between electricity consumption, urbanization and economic growth in Nigeria: fresh insights from Maki Cointegration and causality tests, Heliyon 6 (2) (2020) e03400, http://dx.doi.org/10.1016/j.heliyon.2020.e03400.
- [19] Adedoyin Isola Lawal, Ilhan Ozturk, Ifedolapo O. Olanipekun, et al., Examining the linkages between electricity consumption and economic growth in African economies, Energy 208 (2020) 118363, http://dx.doi.org/10.1016/j. energy.2020.118363.
- [20] Mohammed Bouznit, María P. Pablo-Romero, Antonio Sánchez-Braza, Residential electricity consumption and economic growth in Algeria, Energies 11 (7) (2018) 1656, http://dx.doi.org/10.3390/en11071656.
- [21] Besma Talbi, Mehdi Ben Jebli, Muhammad Farhan Bashir, et al., Does economic progress and electricity price induce electricity demand: A new appraisal in context of Tunisia, J. Public Aff. 22 (1) (2022) e2379, http://dx.doi.org/10. 1002/pa.2379.
- [22] Philip Kofi Adom, Electricity consumption-economic growth nexus: the ghanaian case, Int. J. Energy Econ. Policy 1 (1) (2011) 18.
- [23] Auguste K. Kouakou, Economic growth and electricity consumption in Cote d'Ivoire: Evidence from time series analysis, Energy Policy 39 (6) (2011) 3638–3644.
- [24] Sakiru Adebola Solarin, Electricity consumption and economic growth: Trivariate investigation in botswana with capital formation, Int. J. Energy Econ. Policy 1 (2) (2011) 32–46.
- [25] Godwin Effiong Akpan, Usenobong Friday Akpan, Electricity consumption, carbon emissions and economic growth in Nigeria, Int. J. Energy Econ. Policy 2 (4) (2012) 292.
- [26] Y. Akinwale, O. Jesuleye, W Siyanbola, Empirical analysis of the causal relationship between electricity consumption and economic growth in Nigeria, J. Econ. Manag. Trade (2013) 277–295.
- [27] Adeyemi A. Ogundipe, Ayomide Apata, Electricity consumption and economic growth in Nigeria, J. Bus. Manag. Appl. Econ. 11 (4) (2013).
- [28] Fateh Bélaïd, Fares Abderrahmani, Electricity consumption and economic growth in Algeria: A multivariate causality analysis in the presence of structural change, Energy Policy 55 (2013) 286–295.
- [29] Sakiru Adebola Solarin, Muhammad Shahbaz, Trivariate causality between economic growth, urbanisation and electricity consumption in Angola: Cointegration and causality analysis, Energy Policy 60 (2013) 876–884.
- [30] Yılmaz Bayar, Hasan Alp Ozel, et al., Electricity consumption and economic growth in emerging economies, J. Knowl. Manag. Econ. Inf. Technol. 4 (2) (2014) 1–18.
- [31] Brantley Liddle, Sidney Lung, Might electricity consumption cause urbanization instead ? Evidence from heterogeneous panel long-run causality tests, Global Environ. Change 24 (2014) 42–51.
- [32] Bernard Njindan Lyke, Electricity consumption and economic growth in Nigeria: A revisit of the energy-growth debate, Energy Econ. 51 (2015) 166–176.
- [33] Janneke Dlamini, Mehmet Balcilar, Rangan Gupta, et al., Revisiting the causality between electricity consumption and economic growth in South Africa: a bootstrap rollingwindow approach, Int. J. Econ. Policy Emerg. Econ. 8 (2) (2015) 169–190.
- [34] Augustine C. Osigwe, Damilola Felix Arawomo, Energy consumption, energy prices and economic growth: causal relationships based on error correction model, Int. J. Energy Econ. Policy 5 (2) (2015).
- [35] Nyakundi Michieka, Short-and long-run analysis of factors affecting electricity consumption in sub-saharan africa, Int. J. Energy Econ. Policy 5 (3) (2015) 639–646.
- [36] Bismark Ameyaw, Amos Oppong, Lucille Aba Abruquah, et al., Causality nexus of electricity consumption and economic growth: an empirical evidence from Ghana, Open J. Bus. Manag. 5 (01) (2016) 1.
- [37] Joseph Mawejje, Dorothy N Mawejje, Electricity consumption and sectoral output in Uganda: an empirical investigation, J. Econ. Struct. 5 (1) (2016) 21.
- [38] Muhammad Maladoh Bah, Muhammad Azam, Investigating the relationship between electricity consumption and economic growth: Evidence from South Africa, Renew. Sustain. Energy Rev. 80 (2017) 531–537.

- [40] Dalia M. Ibrahiem, Investigating the causal relationship between electricity consumption and sectoral outputs: evidence from Egypt, Energy Transitions 2 (2018) 31–48.
- [41] Jean Gaston Tamba, Jean Luc Nsouandélé, Armand Fopah Lélé, et al., Electricity consumption and economic growth: evidence from Cameroon, Energy Sources Part B Econ. Plan. Policy 12 (11) (2017) 1007–1014, http://dx.doi.org/10.1080/ 15567249.2017.1349211.
- [42] Benjamin S. Diboma, T. Tamo Tatietse, Power interruption costs to industries in Cameroon, Energy Policy 62 (2013) 582–592.
- [43] Flora Isabelle Métégam Fotsing, Donatien Njomo, Réné Tchinda, Analysis of demand and supply of electrical energy in Cameroon: Influence of meteorological parameters on the monthly power peak of south and north interconnected electricity networks, Int. J. Energy Power Eng. 3 (4) (2014) 168–185.
- [44] SND30, National Development Strategy 2020-2030 for Structural Transformation and Inclusive Development, Republic of Cameroon, pp. 51–82, https://fr. scribd.com/document/492181621/SND30 Strategie Nationale de Deveppement 2020-2030.
- [45] PDSE 2030, Assistance to the Ministry of Energy and Water in the Preparation of the Long Run Development Plan of the Electricity Sector Horizon 2030, Vol. 1, REpublic of Cameroon, Ministry of Energy and Water, 2006, pp. 1–43, https://minepded.gov.cm/2021/09/29/ plandedeveloppementalongtermedusecteurdel'electricitehorizon2030.
- [46] Laura À. Pérez-Sánchez, Raúl Velasco-Fernández, Mario Giampietro, Factors and actions for the sustainability of the residential sector. The nexus of energy, materials, space, and time use, Renew. Sustain. Energy Rev. 161 (2022) 112388.
- [47] Yussra Rashed, Sara Omrani, Danie Nilsson, et al., Sustainability information provision (SIP) framework: A review of the promotion of sustainability in the residential sector, Build. Environ. (2022) 109930.
- [48] Michael Atafo Adabre, Albert P.C. Chan, Critical success factors (CSFs) for sustainable affordable housing, Build. Environ. 156 (2019) 203–214.
- [49] Wendy Miller, Laurie Buys, Factors influencing sustainability outcomes of housing in subtropical Australia, Smart Sustain. Built Environ. 2 (1) (2013) 60-83.
- [50] Yue-Jun Zhang, Hua-Rong Peng, Exploring the direct rebound effect of residential electricity consumption: An empirical study in China, Appl. Energy 196 (2017) 132–141.
- [51] Muhammad Shahbaz, Gazi Salah Uddin, Ijaz Ur Rehman, et al., Industrialization, electricity consumption and CO2 emissions in Bangladesh, Renew. Sustain. Energy Rev. 31 (2014) 575–586.
- [52] Hooi Hooi Lean, Russell Smyth, Electricity consumption and output in ASEAN, Appl. Energy 87 (6) (2010) 1858–1864.
- [53] Eneo Cameroon, Decision n° 0096/arsel/dg/dcec/sdct of 28 2012 setting the electricity sales tariffs (excluding taxes) applicable by eneo cameroun from 2012, 2012, https://eneocameroon.cm/index.php/fr/clients-particuliers-vos-factureset-paiement/clients-particuliers-vos-factures-et-paiement-tarifs-delectricite.
- [54] C. Granados, M. Castaneda, S. Zapata, et al., Feasibility analysis for the integration of solar photovoltaic technology to the Colombian residential sector through system dynamics modeling, Energy Rep. 8 (2022) 2389–2400.
- [55] Reinhard Madlener, Siamak Sheykhha, Wolfgang Briglauer, The electricityand CO2-saving potentials offered by regulation of European video-streaming services, Energy Policy 161 (2022) 112716.
- [56] Eneo Cameroon, Annual report 2020, 2021, https://eneocameroon.cm/pdf/ EneoRapportAnnuel2020.pdf.
- [57] J. Arnau, R. Bendayan, M.J. Blanca, et al., The effect of skewness and kurtosis on the robustness of linear mixed models, Behav. Res. 45 (2013) 873–879, http://dx.doi.org/10.3758/s13428-012-0306-x.
- [58] Angeliki Papana, Catherine Kyrtsou, Dimitris Kugiumtzis, et al., Detecting causality in non-stationary time series using partial symbolic transfer entropy: Evidence in financial data, Comput. Econ. 47 (3) (2016) 341–365, http://dx.doi.org/10. 1007/s10614-015-9491-x.
- [59] Sehrish Gul, Xiang Zou, Che Hashim Hassan, et al., Causal nexus between energy consumption and carbon dioxide emission for Malaysia using maximum entropy bootstrap approach, Environ. Sci. Pollut. Res. 22 (24) (2015) 19773–19785, http://dx.doi.org/10.1007/s11356-015-5185-0.
- [60] Finmath user guide. Augmented dicked-fuller (adf) tests, 2020, https://rtmath. net/assets/docs/finmath.
- [61] Kevin F. Forbes, Ernest M. Zampelli, Wind energy, the price of carbon allowances, and CO2 emissions: Evidence from Ireland, Energy Policy 133 (2019) 110871, http://dx.doi.org/10.1016/j.enpol.2019.07.007.

- [62] John Glynn, Nelson Perera, Reetu Verma, Unit root tests and structural breaks: A survey with applications, 2007, https://ro.uow.edu.au/commpapers.
- [63] Timothy J. Vogelsang, Martin Wagner, A fixed-b perspective on the Phillips-Perron unit root tests, Econom. Theory 29 (3) (2013) 609–628, http://dx.doi. org/10.1017/S0266466612000485.
- [64] Régis Bourbonnais, Économétrie-10e éd.: Cours et Exercices Corrigés, Dunod, 2018.
- [65] Samuel Asumadu-Sarkodie, Phebe Asantewaa Owusu, The relationship between carbon dioxide and agriculture in ghana: a comparison of VECM and ARDL model, Environ. Sci. Pollut. Res. 23 (11) (2016) 10968–10982, http://dx.doi. org/10.1007/s11356-016-6252-x.
- [66] Riitta Hanninen, The law of one price in United Kingdom soft sawnwood importsa cointegration approach. Modern time series analysis in forest products markets, 1999, pp. 55–68, http://dx.doi.org/10.1007/978-94-011-4772-9-4.
- [67] Arif Ullah, Dilawar Khan, Testing environmental Kuznets curve hypothesis in the presence of green revolution: a cointegration analysis for Pakistan, Environ. Sci. Pollut. Res. 27 (10) (2020) 11320–11336.
- [68] Sajjad Barkhordari, Maryam Fattahi, Reform of energy prices, energy intensity and technology: A case study of Iran (ARDL approach), Energy Strategy Rev. 18 (2017) 18–23.
- [69] M.D. Arfanuzzaman, The long-run dynamic relationship between broad money supply and the GDP of Bangladesh: A VECM approach, Dev. Ctry. Stud. 4 (14) (2014) 167–178.
- [70] Djula Borozan, Exploring the relationship between energy consumption and GDP: Evidence from Croatia, Energy Policy 59 (2013) 373–381.
- [71] Juan J. Dolado, Jesus Gonzalo, Francesc Marmol, Cointegration. A Companion to Theoretical Econometrics, 2003, pp. 634–654.
- [72] Walter Enders, Applied Econometric Time Series, John Wiley and Sons, 2008.
- [73] Kwang-Sook Huh, Steel consumption and economic growth in Korea: Longterm and short-term evidence, Resour. Policy 36 (2) (2011) 107–113, http: //dx.doi.org/10.1016/j.resourpol.2011.01.005.
- [74] Nahla Samargandi, Kazi Sohag, The interaction of finance and innovation for low carbon economy: Evidence from Saudi Arabia, Energy Strategy Rev. 41 (2022) 100847.
- [75] Muhammad Mohsin, Sobia Naseem, Muhammad Zia-Ur-Rehman, et al., The crypto-trade volume, GDP, energy use, and environmental degradation sustainability: An analysis of the top 20 crypto-trader countries, Int. J. Finance Econ. 28 (1) (2023) 651–667.
- [76] Muhammad Aslam, Using heteroscedasticity-consistent standard errors for the linear regression model with correlated regressors, Comm. Statist. Simulation Comput. 43 (10) (2014) 2353–2373.
- [77] Régis Bourbonnais, Sophie Méritet, Electricity spot price modelling: univariate time series approach, in: The Econometrics of Energy Systems, Palgrave Macmillan, London, 2007, pp. 51–74, http://dx.doi.org/10.1057/ 97802306263173.
- [78] Godfrey K. Putunoi, Cyrus M. Mutuku, Domestic debt and economic growth nexus in Kenya, Curr. Res. J. Econ. Theory 5 (1) (2013) 1–10, http://dx.doi. org/10.9790/5933-06230914.
- [79] Agumas Alamirew Mebratu, Mebratu-B-PLC theory implication on developing country's oxygen of bank, Cogent Econ. Finance 11 (1) (2023) 2166210.
- [80] Biu Emmanuel, Nwakuya T Maureen, Nduka Wonu, Detection of non-normality in data sets and comparison between different normality tests, Asian J. Probab. Stat. 5 (4) (2020) 1–20.
- [81] A. Enisan Akinlo, The stability of money demand in Nigeria: An autoregressive distributed lag approach, J. Policy Model. 28 (4) (2006) 445–452.
- [82] Sakib Bin Amin, Foqoruddin Al Kabir, Farhan Khan, Energy-output nexus in Bangladesh: A two-sector model analysis, Energy Strategy Rev. 32 (2020) 100566, http://dx.doi.org/10.1016/j.esr.2020.100566.
- [83] L. Al-Labadi, M. Zarepour, Two-sample Kolmogorov–Smirnov test using a Bayesian nonparametric approach, Math. Methods Statist. 26 (3) (2017) 212–225, http://dx.doi.org/10.3103/S1066530717030048.
- [84] Kwaku Addai, Berna Serener, Dervis Kirikkaleli, Empirical analysis of the relationship among urbanization, economic growth and ecological footprint: Evidence from eastern europe, Environ. Sci. Pollut. Res. 29 (19) (2022) 27749–27760.
- [85] Yi-Tui Chen, The factors affecting electricity consumption and the consumption characteristics in the residential sector-a case example of Taiwan, Sustainability 9 (8) (2017) 1484.
- [86] Bright Akwasia Gyamfi, Murad A. Bein, Festus Victor Bekun, Investigating the nexus between hydroelectricity energy, renewable energy, nonrenewable energy consumption on output: evidence from E7 countries, Environ. Sci. Pollut. Res. 27 (2020) 25327–25339.